Author's response to reviews

Title: Different atrophy-hypertrophy transcription pathways in muscles affected by severe and mild spinal muscular atrophy

Authors:

Caterina Millino (caterina@cribi.unipd.it)
Marina Fanin (marina.fanin@unipd.it)
Andrea Vettori (andrea.vettori@unipd.it)
Paolo Laveder (paolo.laveder@unipd.it)
Maria Luisa Mostacciulo (marialuisa.mostacciulo@unipd.it)
Corrado Angelini (corrado.angelini@unipd.it)
Gerolamo Lanfranchi (gerolamo.lanfranchi@unipd.it)

Version: 2 Date: 25 February 2009

Author's response to reviews:

Dr. Robin Cassady-Cain
In-house Editor
BMC Medicine

Dear Dr. Cassady-Cain,

I am sending you the manuscript entitled: "Different atrophy-hypertrophy transcription pathways in muscles affected by severe and mild spinal muscular atrophy" by Millino et al for publication in BMC Medicine.

This manuscript has been originally accepted for publication in BMC Medical Genomics, but I have agreed with the kind invitation of Dr. Scott Edmunds, Senior Editor of BMC-series Journals, who asked me to move it to BMC Medicine.

Therefore I hope that you could accept our paper for publication in BMC Medicine.

Yours sincerely,

Gerolamo Lanfranchi

Professor of Genetics
CRIBI Biotechnology Centre and Department of Biology
University of Padova, Padova, Italy